Author's response to reviews

Title: Evidence that an APOE epsilon4 "double whammy" increases risk for Alzheimer's disease

Authors:

  Ina Caesar (ina.caesar@exchange.mssm.edu)  
  Sam Gandy (samuel.gandy@mssm.edu)

Version: 6 Date: 2 April 2012

Author's response to reviews:

Final revisions as per journal editors made here.